January 19, 2017

Mr. Lian-Sheng Ma  
President and Company Editor-in-Chief  
Baishideng Publishing Group, Inc.  
8226 Regency Drive  
Pleasanton, CA 94588

Dear Mr. Ma,

Thank you for accepting our review article titled Pathogenesis and Clinical Spectrum of Primary Sclerosing Cholangitis to the *World Journal of Gastroenterology*. The following reviewer comments have been addressed in our revised article:

I. “You’d better offer us one paper with word format next (such as, .doc or .docx), which is easy to be edited and helpful to accept earlier. Thank you very much!”

   a) Please accept our sincerest apologizes for this oversight. The revised manuscript has been uploaded in a .doc format for your review.

II. “Name of journal: World Journal of Gastroenterology, ESPS Manuscript NO, Manuscript type.”

   a) These requested changes have been incorporated within the reviewed manuscript.

III. “Title: Title should not appear in abbreviation, please correct it, also should not be more than 12 words and including no less than 2 key words. Running title: (Less than 6 words).”

   a) Our revised manuscript has incorporated these revisions.

IV. “Please rearrange all the authors’ affiliations with Department, University or Institute, City, Postcode, Country, etc. (without any symbol or figure like * or 1, postcode must be there) such as: full name, address.”

   a) Our revised manuscript has incorporated these revisions.

V. “Author contribution.”

   a) Our revised manuscript has incorporated these revisions.
VI. “Conflict-of-interest statement.”
   a) Our revised manuscript has incorporated this revision.

VII. “Correspondence to: Telephone: Fax:”
   a) Our revised manuscript has incorporated this revision.

VIII. “Abstract, no less than 200 words.”
   a) Our revised manuscript has incorporated this revision.

IX. “Key words: 5-10 words.”
   a) Our revised manuscript has incorporated this revision.

X. “Core tip. Audio core tip.”
   a) Our revised manuscript has incorporated these revisions. The audio core tip has been uploaded as a separate file, as per your instructions.

XI. “Please provide all authors abbreviation names and manuscript title here. World J Gastroenterol 2016; In press.”
   a) Our revised manuscript has incorporated this revision.

XII. “Conclusion.”
   a) Our revised manuscript has incorporated this revision.

XIII. “Please check that there are no repeated references! Please add PubMed citation numbers and DOI citation to the reference list and list all authors.”
   a) All references have been reviewed, and there are no duplicate references within the manuscript.
   b) The PubMed citation has been added to each reference. The DOI citation has also been incorporated, when this is available for the respective reference.
   c) All authors have been added to their respective reference.
The revised review article has been uploaded online in conjunction with other requested documents, per the instructions provided in your previous correspondence.

Thank you for your time and I look forward to hearing from you.

Sincerely,

Kiron M. Das, M.D., Ph.D.

Corresponding Author